

A coupled cloud physics-radiation parameterization of the bulk optical properties of cirrus and its impact on the Met Office unified model global atmosphere 5.0 configuration

Article

Published Version

Baran, A. J., Hill, P. ORCID: https://orcid.org/0000-0002-9745-2120, Furtado, K., Field, P. and Manners, J. (2014) A coupled cloud physics-radiation parameterization of the bulk optical properties of cirrus and its impact on the Met Office unified model global atmosphere 5.0 configuration. Journal of Climate, 27 (20). pp. 7725-7752. ISSN 1520-0442 doi: https://doi.org/10.1175/JCLI-D-13-00700.1 Available at https://centaur.reading.ac.uk/40191/

It is advisable to refer to the publisher's version if you intend to cite from the work. See <u>Guidance on citing</u>.

To link to this article DOI: http://dx.doi.org/10.1175/JCLI-D-13-00700.1

Publisher: American Meteorological Society

All outputs in CentAUR are protected by Intellectual Property Rights law, including copyright law. Copyright and IPR is retained by the creators or other copyright holders. Terms and conditions for use of this material are defined in



the End User Agreement.

www.reading.ac.uk/centaur

CentAUR

Central Archive at the University of Reading

Reading's research outputs online

A Coupled Cloud Physics–Radiation Parameterization of the Bulk Optical Properties of Cirrus and Its Impact on the Met Office Unified Model Global Atmosphere 5.0 Configuration

ANTHONY J. BARAN

Met Office, Exeter, United Kingdom

PETER HILL

Met Office, Reading, United Kingdom

KALLI FURTADO, PAUL FIELD, AND JAMES MANNERS

Met Office, Exeter, United Kingdom

(Manuscript received 13 November 2013, in final form 6 June 2014)

ABSTRACT

A new coupled cloud physics-radiation parameterization of the bulk optical properties of ice clouds is presented. The parameterization is consistent with assumptions in the cloud physics scheme regarding particle size distributions (PSDs) and mass-dimensional relationships. The parameterization is based on a weighted ice crystal habit mixture model, and its bulk optical properties are parameterized as simple functions of wavelength and ice water content (IWC). This approach directly couples IWC to the bulk optical properties, negating the need for diagnosed variables, such as the ice crystal effective dimension. The parameterization is implemented into the Met Office Unified Model Global Atmosphere 5.0 (GA5) configuration. The GA5 configuration is used to simulate the annual 20-yr shortwave (SW) and longwave (LW) fluxes at the top of the atmosphere (TOA), as well as the temperature structure of the atmosphere, under various microphysical assumptions. The coupled parameterization is directly compared against the current operational radiation parameterization, while maintaining the same cloud physics assumptions. In this experiment, the impacts of the two parameterizations on the SW and LW radiative effects at TOA are also investigated and compared against observations. The 20-yr simulations are compared against the latest observations of the atmospheric temperature and radiative fluxes at TOA. The comparisons demonstrate that the choice of PSD and the assumed ice crystal shape distribution are as important as each other. Moreover, the consistent radiation parameterization removes a long-standing tropical troposphere cold temperature bias but slightly warms the southern midlatitudes by about 0.5 K.

1. Introduction

The Fifth Assessment Report of the Intergovernmental Panel on Climate Change (IPCC; Stocker et al. 2013) concluded that the radiative coupling between clouds and the atmosphere remains one of the largest sources of uncertainty in understanding intermodel differences in predicting the equilibrium state of Earth's climate. Stocker et al. (2013) concluded that most of the intermodel

E-mail: anthony.baran@metoffice.gov.uk

differences were due to low clouds. It is well known, however, that general circulation models (GCM) generally underpredict ice mass, compared to observations, by several or more factors (Waliser et al. 2009; Wu et al. 2009; Delanoë et al. 2011; Field et al. 2014). This general underprediction of high-troposphere ice mass by climate models creates problems when assessing the importance of different high cloud types to the climate sensitivity of GCMs. This is because the role of high clouds in determining the shortwave (SW) and longwave (LW) radiative exchanges may well be underestimated by those models (Baran 2012). Moreover, the parameterization of high clouds can affect the amount of low clouds predicted by the model through the vertical profile of

Corresponding author address: Dr. Anthony J. Baran, Met Office, Cordouan 2, FitzRoy Road, Exeter, Devon EX1 3PB, United Kingdom.

radiative heating in the model, as shown by McFarquhar et al. (2003). To quantify the role of the ice cloud in the radiative coupling between the atmosphere and cloud, it is of primary importance to construct accurate parameterizations of its bulk optical properties. Unfortunately, this is currently far from being achieved, as different ice crystal models and parameterizations can lead to significantly different GCM-simulated tropical LW and SW flux differences at the top of the atmosphere (TOA) of between about 10 and 30 W m^{-2} and $-10 \text{ and } -35 \text{ W m}^{-2}$, respectively (Baran 2012). The study by Fu (2007) showed that changing the aspect ratio of hexagonal ice columns from 1.0 to 0.1 leads to a shortwave cloud radiative effect (CRE) difference of about $-40 \,\mathrm{Wm^{-2}}$ at TOA. The Fu (2007) result was obtained by assuming a completely overcast sky and an optical depth of 4.0. In general, a recent study by Williams and Webb (2009) showed that the divergence between different climate models in predicting the CRE of cirrus can be $\pm 20 \,\mathrm{W \,m^{-2}}$ and between -50 and 15 Wm^{-2} , in the tropics and extratropics, respectively.

However, the differences highlighted above could be due to different cloud physics parameterizations, such as particle size distributions (PSDs) and/or ice optics, being assumed in different climate models. The study by Edwards et al. (2007, hereafter E07) showed that it is possible to obtain insignificant differences in the simulated SW and LW fluxes at TOA between seemingly different ice crystal optical parameterizations if the PSD and shape of ice crystals are known and the same assumptions are made between the ice crystal size and environmental temperature. One further advance on this work would be to assume the same mass-dimensional relationship and PSD in the cloud physics and radiation schemes of the climate model. A second advance on the previous work would be to insist that the ice crystal model assumed in the radiation scheme must follow the currently observed area and mass-dimensional relationships that have been derived from in situ measurements (Mitchell et al. 2008; Baran 2009, 2012; Cotton et al. 2013).

The studies of (Baran and Labonnote 2007; Mitchell et al. 2008; Baran et al. 2009, 2011a,b, 2014a) have shown if the idealized ice crystal model follows observed area and mass–dimensional relationships, then the need for assumptions about relationships between an effective dimension of the ice crystal population (the effective dimension is defined below) and environmental temperature can be negated when predicting the radiative properties of ice cloud. The approach of the previous authors means that microphysical assumptions in both the cloud physics and radiation schemes in the climate model are consistent, whereas optical property parameterizations that rely on assumed relationships between an effective dimension and environmental temperature tend to be physically inconsistent between the cloud physics and radiation schemes (Baran 2012). This physical inconsistency arises because different assumptions are made between the cloud physics and radiation schemes with regard to shape of the PSD and the area and mass-dimensional relationships. In general, current parameterizations of ice crystal optical properties tend to be functions of ice crystal effective dimension, ice water content (IWC), and/or environmental temperature.

The ice crystal effective dimension D_e is defined as the ratio between the IWC and the volume extinction coefficient, where the volume extinction coefficient is the sum of the scattering and absorption coefficients (Francis et al. 1994, 1995). However, there are multiple definitions of D_e that can be employed, as shown by McFarquhar and Heymsfield (1998), and these definitions can vary in their values by much more than 20%. For some definitions of D_e , it is itself an optical property that can be used to describe the extinction of light in a cloud. This is because it is inversely related to the mass extinction coefficient (Mitchell et al. 1996; McFarquhar and Heymsfield 1998; Wyser and Yang 1998; Kokhanovsky 2004; Mitchell 2002). This inverse relationship is only valid for solar wavelengths, because at those wavelengths, where the wavelength of the incident solar radiation is much smaller than the largest dimension of the ice crystal, the extinction coefficient is twice the ice crystal orientation-averaged projected area (van de Hulst 1957). Therefore, given D_e and IWC, the radiative properties of cirrus can be uniquely determined for solar wavelengths under the assumption of a homogeneous distribution of IWC within a single layer.

Most of the current parameterizations of cirrus bulk optical properties used in climate models depend on empirically derived deterministic relationships between D_e and environmental temperature and/or IWC (Fu et al. 1999; Kristjánsson et al. 2000; E07; Bozzo et al. 2008; Hong et al. 2009; Gu et al. 2011; Yi et al. 2013). Generally, there is no evidence for exact deterministic relationships between D_e and environmental variables and/or IWC. In the studies reported by Lynch et al. (2002) there appear to be very weak correlations between D_e and environmental variables, such as in-cloud temperature. A further limitation of some of the earlier parameterizations cited above and the lack of correlation between D_e and in-cloud temperature and/or IWC, is that the historical in situ-measured PSDs on which they depend are now known to be biased toward populations of small ice crystals. This is due to the shattering of ice crystals on the housing and/or at the inlets of the microphysical probes that have been used to measure the size spectra (Field et al. 2003, 2006; Korolev et al.

2011, 2013). By mitigating the effects of shattering, more recent studies by Mitchell et al. (2011b) have demonstrated stronger correlations between D_e , in-cloud temperature, and IWC.

How the issue of shattering affects previous cirrus radiative parameterizations depends on the definition of D_e used in those parameterizations. For instance, it can be seen in Korolev et al. (2013) that shattering on microphysical probes affects, for particles of size less than about $500 \,\mu\text{m}$ and in the temperature range from 0° to -35°C , both the extinction and IWC in an approximately similar way (this is because extinction and IWC may have a similar dependence on the diameter of the ice crystals). The effect of shattering on the extinction and IWC is to increase them by, on average, 20% and 30%, respectively (Korolev et al. 2013). Therefore, the effect of shattering on their ratio will be small but may not completely cancel the effect on estimates of D_e . However, if D_e is defined as the ratio of the third moment (i.e., volume) to the second moment (i.e., area) of the PSD, then shattering will bias the calculated value of D_e to smaller values. The latter definition of D_e is the one that is assumed in the current Met Office operational Global Atmosphere model, and the PSDs used in the current Met Office parameterization were applied without removing any shattered ice crystal artifacts. In the study by Mitchell et al. (2010), it is shown that shattering could cause the in situ–estimated D_e to be underpredicted by several factors, and in that paper for ice crystal size less than 240 μ m, the mass of ice tends to that of a sphere. The impact of shattering on IWC and extinction is still uncertain, as it does depend on the extent of the PSD and on what size of ice crystal contributes most to the IWC. Clearly, further research in this area is required to determine the small ice mode (Korolev et al. 2013). A possible consequence of this underprediction of D_e is to bias the averaged mass extinction of clouds to higher values, which means that the reflected SW radiation at TOA is biased to larger values (since mass extinction is inversely related to D_e). Moreover, in the SW, the underprediction of IWC in most GCMs leads to darker clouds, while the bias toward small D_e leads to brighter clouds. Hence, the two errors effectively cancel, which may lead to about the right emergent SW fluxes being predicted in the GCM. This cancellation of SW error excludes the possibility of exposing systematic climate model errors by comparing model SW flux predictions directly against SW flux measurements. However, there are large uncertainties in a global climate model, so this SW effect may not be apparent. It is more difficult to state the impact of the parameterization error on the outgoing LW flux, as that fundamentally depends on where the model locates the cloud in the atmosphere. For instance, if the model predicts the cirrus to be too optically thin and not high enough in the atmosphere, then this would increase the outgoing LW flux, and the overestimate of mass extinction may not necessarily be sufficient to cancel the error. To assess the performance of models in the SW and LW, it may prove more useful to move away from flux comparisons to high-resolution radiance comparisons, as suggested by Goody et al. (1998). However, to test model physical consistency across the electromagnetic spectrum, SW and LW high-resolution radiance measurements need to be combined (Baran and Francis 2004).

Yet a further difficulty with previous parameterizations is that, generally, the assumed PSDs used in the cloud physics and radiation schemes are not consistent. This means that D_e -based radiative parameterizations result in different D_e value being predicted in the radiation scheme, while in the cloud physics scheme, for the same mass of ice, a different D_e could be predicted. If so, this is physically inconsistent and, as such, should be removed from climate models. Note that the inconsistent types of parameterizations discussed above are prevalent in climate models that are used in the IPCC series of reports.

For D_e to be generally useful, it must be applicable across the spectrum. That is, the radiative properties of ice cloud must be uniquely determined from D_e and IWC alone. Unfortunately, this is not generally found to be the case. At infrared and radar wavelengths, the concept of D_e is no longer valid, as demonstrated and discussed by several authors (Mitchell 2002; Baran 2005, 2007; De Leon and Haigh 2007; Mitchell et al. 2011a; Baran et al. 2011b). The reason for the breakdown in the validity of the D_e concept at infrared wavelengths is simply because, at those wavelengths, the ice crystal size becomes small relative to the incident wavelength, which means that the mass extinction coefficient is no longer inversely related to D_{e} (Baran 2005). More fundamentally, as the ice crystals become small relative to the incident wavelength at infrared wavelengths, the physical process of absorption through electromagnetic wave resonance effects becomes more important, so absorption by ice is no longer uniquely determined by the ice crystal volume or projected area (Mitchell 2002; Mitchell et al. 2011a). At radar wavelengths, the radar reflectivity becomes more sensitive to aggregated ice crystals. As new ice crystals grow in size by vapor diffusion and also become aggregated, with the larger ice crystals more likely to aggregate, the ice crystal massdimensional relationship asymptotes to D^2 , where D is the ice crystal maximum dimension (Westbrook et al. 2004; Heymsfield et al. 2010; Cotton et al. 2013). As a result of this asymptotic behavior in ice mass, the

dependence of the numerator in the D_e formula tends to D^2 , while the denominator is also proportional to D^2 ; hence D_e in regions of ice crystal aggregation, tends to constant values. This behavior in D_e can be seen in some of the profiled aircraft observations of Francis (1995), which show little vertical variation in effective radius (i.e., effective radius = $D_e/2$), and it is noted by Francis (1995) that the lack of variation in effective radius was due to the similar dependence that mass and area have on ice crystal size. This same behavior was also noted by McFarquhar and Heymsfield (1998). Clearly, in this situation, D_e tends to a constant value with respect to the ratio of IWC to extinction. Moreover, because of the nonuniqueness of cirrus PSDs, there is the possibility that there is no unique radiative solution for a particular value of D_e (Mitchell 2002).

In this paper, an alternative to the D_e approach is presented for the parameterization of the bulk optical properties of ice cloud. That is, the bulk optical properties are directly linked to the climate model prediction of IWC. This alternative approach enables the radiative properties of ice clouds to be directly related to a GCM prognostic variable rather than some generally diagnosed quantity, such as D_e . A similar approach was used by McFarquhar et al. (2003), where the bulk optical properties were equivalently parameterized as a function of IWC.

The climate model used as a basis for the simulations in this paper is the Met Office Unified Model Global Atmosphere 5.0 (GA5) configuration. The new parameterization is tested by comparing the results of the climate model simulations against the Clouds and the Earth's Radiant Energy System (CERES) reanalyzed product of the annual 20-yr globally averaged reflected and outgoing longwave (OLR) fluxes at TOA. The CERES product used here is described in Loeb et al. (2009). The parameterization proposed by E07, a D_e -based scheme, is also compared against the same CERES products under the same microphysical assumptions as used in the experiments using the new parameterization. The impacts of all radiation parameterizations on the temperature structure of the troposphere are also discussed and are compared against the Interim European Centre for Medium-Range Weather Forecasts (ECMWF) Re-Analysis (ERA-Interim) of atmospheric temperature observations, covering the period 1989–2008 (Dee et al. 2011).

The paper is split into the following sections. Section 2 briefly describes the GA5 configuration, and the atmospheric parameterizations within it that are most pertinent to this paper. Section 3 briefly describes the self-consistent scattering model for cirrus and describes the bulk optical properties that the model predicts. Section 4 presents the parameterization along with an

error analysis of the parameterization, and section 5 describes the results of the various experiments. Section 6 presents the conclusions and discusses the most important points of this paper.

2. The GA5 configuration and atmospheric parameterizations

The approach of the Met Office in simulating the weather and climate is to keep the atmospheric physics parameterizations used for both the same, as far as possible, for use on all time scales (Brown et al. 2012). This has led to an annual definition of a Global Atmosphere (GA) configuration of the Met Office Unified Model, and this definition of a single GA model has been described by Walters et al. (2011). In this paper, use is made of the GA5 configuration, which is based on the GA4 configuration (Walters et al. 2013) but includes a new dynamical core. The new dynamical core is fully described by Wood et al. (2014). The details of the atmospheric parameterizations used in the GA5 configuration can be generally found in Walters et al. (2013). The GA5 configuration used in this study consists of 85 levels, the highest level being at an altitude of 85 km, and all the simulations used in this study are run at a horizontal resolution of about 135 km in the midlatitudes.

However, the GA parameterizations that are particularly pertinent to this paper are briefly discussed here. The radiative transfer model used in the GA series of models is the Edwards and Slingo (1996) scheme. Gaseous absorption is treated within the radiative transfer model by using the correlated-k method with six bands in the shortwave and nine bands in the longwave. In the longwave and shortwave, multiple scattering is fully included in the Edwards and Slingo (1996) scheme. A further difference between GA4 and GA5 is that, in GA4, a full radiative transfer calculation is done every three hours, and a fast calculation is done every hour using a few k terms to account for cloud changes, as described in Manners et al. (2009), whereas, in the GA5 configuration, a full radiative transfer calculation is done every hour. For a more detailed description of the radiation scheme, see section 2.2 of Walters et al. (2013), and the cloud scheme is fully described in sections 2.3-2.4 and in 3.3–3.4 of that paper. The GA convection scheme is described in section 3.6 of Walters et al. (2013).

The current GA operational ice optics parameterization is described by E07. The bulk ice optical parameterization developed in that paper is based on the 8-monomer hexagonal ice crystal model developed by Yang and Liou (1998). The hexagonal ice aggregate model is invariant with respect to size, so the asymmetry parameter will also remain invariant with respect to size. Observations generally show that aggregates of ice crystals grow longitudinally rather than radially, so their area ratios (i.e., the ratio of the projected area of a nonspherical particle to the area of the circumscribing circle of the same maximum dimension as the nonspherical particle) decrease as a function of increasing ice crystal size (Heymsfield and Miloshevich 2003; Field et al. 2008; McFarquhar et al. 2013). The asymmetry parameter is naturally related to the ice crystal size; so it is also related to their area ratio or aspect ratio (Fu 2007); therefore, this parameter should change with ice crystal size. Furthermore, observations used by Baum et al. (2005) and Baran et al. (2011b) show that the hexagonal ice aggregate model overpredicts in situderived IWC by several factors. The hexagonal ice aggregate shortwave and longwave bulk optical properties used in the E07 parameterization were compiled by Baran and Francis (2004). To address the physical inconsistencies inherent in the hexagonal ice aggregate model discussed above, an ensemble model of ice crystals developed by Baran and Labonnote (2007) has been proposed, and this model is briefly described in section 3 of this paper.

The assumed ice PSD that is currently used in GA5 is a parameterization developed by Houze et al. (1979); it is based on 37 in situ PSDs, and 90% of these were measured at temperatures warmer than -30° C. Currently, the Houze et al. (1979) estimated PSD is kept constant at temperatures colder than -35° C. This assumption means that at much colder temperatures, there will be orders of magnitude more frequently occurring large ice crystals than there should be. This has clear implications for the assumed fall speed, which must be artificially increased to accommodate space-based radiometric SW and OLR measurements.

A more recent parameterization of the ice PSD has been proposed by Field et al. (2007, hereafter F07). The parameterization of F07 is a moment estimation scheme, and it is based on 10000 in situ PSDs that were measured between the temperatures of about 0° and -60° C in the midlatitudes and tropics. Figure 1 shows the measurement space in IWC and cloud temperature over which the F07 and Houze et al. (1979) parameterizations are based. The figure shows that the F07 PSD parameterization is based on measurements over a far greater range of IWC and cloud temperature than the Houze et al. (1979) parameterization. As previously stated, the historical PSDs have been affected by ice crystal shattering. The technique of filtering employed by F07 to reduce the effect of shattering is described in Field et al. (2006) and so will not be repeated here. The F07 parameterization also ignores all in situ measurements of ice crystals with maximum dimensions less than $100 \,\mu\text{m}$. For ice crystals less than $100\,\mu\text{m}$ in size, the F07 parameterization assumes an exponential distribution (Field and Heymsfield 2003), which is added to a modified gamma distribution at an ice crystal size of about $100 \,\mu$ m. As previously discussed, shattered ice crystal artifacts may still be present for ice crystals less than $500 \,\mu\text{m}$ in size (Korolev et al. 2011, 2013). However, the extent to which the F07 parameterization has been affected by shattering is unclear. In Korolev et al. (2013) it is shown that if filtering alone is applied to remove shattered artifacts (Field et al. 2003, 2006), then significant divergence from the best estimate of the PSD does not occur until ice crystal size is less than about 200–175 μ m. It cannot be categorically stated that the F07 parameterization has not been affected by shattering, and if so, by how much, since the parameterization is based on in situ data that covers a far greater temperature and altitude range than was considered in Korolev et al. (2013). However, the F07 parameterization is still a more representative parameterization of the PSD than Houze et al. (1979), and the latter parameterization did not employ any techniques to try to remove shattered ice crystal artifacts from their parameterization. Moreover, Furtado et al. (2014) show that the F07 parameterization is a better fit to in situ-derived higher moments of the PSD than the Houze et al. (1979) parameterization. The in situ data used by Furtado et al. (2014) is based on microphysical probes, which have the necessary modifications attached to them to reduce shattering, and filtering has been applied to further reduce the effects of shattering on the moments of the PSD (Field et al. 2003, 2006; Cotton et al. 2013; Korolev et al. 2011, 2013).

The moments in the F07 parameterization are predicted as a function of IWC and cloud temperature, where the moment carrying ice mass is generally related to the IWC, and all other moments are generated from the second moment and cloud temperature. To generate the PSD from the F07 moment estimation parameterization, the second moment is related to an assumed mass-dimensional relationship.

In GA5, the Brown and Francis (1995) derived massdimensional relationship is assumed, where the mass of each ice crystal is given by $0.0185D^{1.9}$, where *D* is the ice crystal maximum dimension (both mass and *D* are in SI units, which are used throughout this paper). However, a recent midlatitude cirrus campaign conducted by Cotton et al. 2013 found that the Brown and Francis (1995) relationship overestimated in situ bulk measurements of IWC by, at most, a factor of 2. In the same paper, the authors found that the mass-dimensional relationship that best fitted the measurement of bulk



FIG. 1. The IWC and cloud temperature measurement space over which the F07 (light gray shaded area) and Houze et al. (1979) (dark gray shaded area) PSD parameterizations are based. The contours are the mass-weighted mean size as a function of IWC and in-cloud temperature and range from a value of $50 \,\mu\text{m}$ at the bottom lower left to $2750 \,\mu\text{m}$ at the top right.

IWC was found to be 0.0257D². Moreover, the massdimensional relationship derived by Cotton et al. 2013 is also consistent with the mass-dimensional relationship found independently by Heymsfield et al. (2010) using bulk measurements of IWC obtained in midlatitude and tropical cirrus. The ice microphysics assumed in the model experiments used in this paper are based on the F07 PSD parameterization, the mass-dimensional relationship derived by Cotton et al. 2013, and a fall speed parameterization that has been developed by Furtado et al. (2014). In the next section, the ensemble model of Baran and Labonnote (2007) is described, along with its bulk optical properties.

3. The ensemble model and its single-scattering properties

It has recently been argued by Baran (2009, 2012) that ice optical properties should be directly linked to global model prognostic variables, such as IWC. It was also argued by the same author that the same PSDs should be assumed in both the radiation and cloud physics schemes within climate models. To this end, the ensemble model of cirrus ice crystals was developed to predict the orientation-averaged cross sections of naturally occurring ice crystals and conserve ice crystal mass (Baran and Labonnote 2007). The ensemble model of cirrus ice crystals is shown in Fig. 2, and the figure shows that the ensemble model consists of six elements, the first of which is the hexagonal ice column of aspect ratio unity; the second element is the six-branched bullet rosette. Thereafter; the ensemble model consists of hexagonal ice aggregates forming 3-, 5-, 8-, and 10-monomer ice aggregates. Each of the ice aggregates is constructed by arbitrarily attaching hexagonal monomers to each other until an ice aggregate is constructed. The individual monomers are sufficiently spaced from each other to minimize multiple reflections between monomers. The first element represents the smaller sizes of ice crystals in the PSD, while the hexagonal ice aggregates represent the process of ice crystal aggregation and,



FIG. 2. The Baran and Labonnote (2007) ensemble model of cirrus ice crystals as a function of ice crystal maximum dimension D_{max} . (top) The first 3 members of the ensemble model, which are the hexagonal ice column, 6-branched bullet rosette, and a 3-monomer ice aggregate. (bottom) The next three ensemble members, which are the 5-, 8-, and 10-monomer ice aggregates.

thus, represent the larger sizes of ice crystals in the PSD. The members of the ensemble are distributed into six equal size intervals of the F07 PSD. This distribution means that, for radiative transfer in the solar and thermal regions of the spectrum, the cloud extinction is chiefly determined by the first element, with no contributions from the fourth member onward: however, for microwave and radar calculations, the cloud extinction would be weighted toward the more aggregated ensemble members (i.e., fourth member onward). For the same ice crystal maximum dimension, the more aggregated ensemble members would have much lower values of orientation-averaged cross sections than the first member of the ensemble model. The weightings critically depend on the shape of the PSD, and the weightings could change, given different shapes of the PSD. The impact of changing the weighting of the ensemble model on the reflected shortwave and OLR at TOA and on the temperature structure of the troposphere, is explored in one of the experiments contained in this study.

With the distribution of ensemble members described above, a series of papers by Baran et al. (2009, 2011a,b, 2014a) demonstrated that the model could predict, to within current experimental uncertainties, the solar volume extinction coefficient and IWC of midlatitude and tropical cirrus. More recently Baran et al. (2014b) demonstrated that the ensemble model prediction of the area ratio was within the range found by a number of independent authors, who used in situ estimations of the area ratio obtained in midlatitude, tropical, and Arctic cirrus (McFarquhar et al. 2013; Field et al. 2008; Heymsfield and Miloshevich 2003).

Other habit mixture models have also been proposed, such as those of Baum et al. (2005, 2011), that incorporate other observed habits not considered here, such as hexagonal ice plates, aggregates of plates, hollow columns, and hollow bullet rosettes.

Moreover, the ensemble model has been demonstrated to be physically consistent across the electromagnetic spectrum, from the UV to the radar frequency of 35 GHz, using consistent microphysics across the spectrum (Baran et al. 2011b, 2014a). Using measurements from the UV-to-radar frequencies, it was shown by Baran et al. (2014a) that for one case of optically thin midlatitude cirrus (i.e., solar optical depth $\ll 1$), the ensemble model could predict aircraft-mounted backscatter lidar-derived UV volume extinction coefficients to generally within $\pm 25\%$ for all altitudes considered. Moreover, it was shown that aircraft-based high-resolution brightness temperature measurements at wavelengths between about 8.0 and $12.0 \,\mu\text{m}$ and between about 3.3 and 5.0 μ m could be simulated to within ± 1 and ± 2 K of the measured brightness temperatures, respectively. Therefore, the optical properties used in the parameterization presented in this paper have been demonstrated to be physically consistent across those parts of the electromagnetic spectrum of relevance to climate models. Furthermore, a recent global study by Vidot et al. (2014, manuscript submitted to J. Geophys. Res.), demonstrated that the ensemble model ice optical properties produced a mean departure of only 0.43 K from space-based brightness temperature measurements. The measurements were located in the terrestrial window region at 8.0, 11.0, and $12.0 \,\mu$ m. This study was comprised of 26791vertical profiles of cirrus, retrieved from radar and lidar profiles, with visible optical depths ranging from 0.03 to 4.0, and these profiles were distributed between the latitudes of about 70°N and 60°S, and at altitudes from about 440 to 50 hPa. Therefore, given the range of cirrus used in the study by Vidot et al. (2014, manuscript submitted to J. Geophys. Res.), the ice optical properties used here can be applied to a climate model.

The consistent microphysics used in Baran et al. (2014a) was derived from the ensemble model, and the derivation is described in Baran et al. (2011b). In that paper, using

observational data of ice crystal mass obtained in deep tropical convection by Field et al. (2008), the ensemble model was shown to fit a mass–dimensional relationship of the form $0.04D^2$. This mass–dimensional relationship was shown by Cotton et al. (2013) to be within the experimental uncertainty of other independently derived relationships, which were based on in situ bulk measurements of IWC obtained in a variety of cirrus. In this study, the ensemble model mass–dimensional relationship is used to generate the F07 PSDs, and the impacts of these PSDs on the reflected shortwave and OLR at TOA, as well as on the temperature structure of the troposphere, are discussed in a later section of the paper.

To simulate the SW and LW fluxes in the climate model the total optical properties are required, which are the bulk extinction coefficient β_{ext} , bulk scattering coefficient β_{sca} , single-scattering albedo ω_0 , and the asymmetry parameter g. These are wavelength dependent, but this dependency has been dropped here for reasons of clarity. The methods used to compute the total optical properties predicted by the ensemble model have already been described by Baran et al. (2014a) and will not be repeated here for reasons of brevity. The bulk optical property $\beta_{ext/sca}$ is defined by the following equation:

$$\beta_{\text{ext/sca}} = \int C_{\text{ext/sca}}(\bar{l})n(\bar{l})\,d\bar{l}\,,\tag{1}$$

where $C_{\text{ext/sca}}(\bar{l})$ are the orientation-averaged scattering and extinction cross sections of each of the elements of the ensemble model, the vector I contains each ensemble model element as a function of its size, and $n(\bar{l})$ is given by the F07 moment estimation parameterization. Therefore, the bulk extinction and scattering coefficients can be expressed as functions of IWC and temperature, since these two parameters are used to generate the F07 PSDs. An alternative to this, if one wanted to relate the bulk optical properties to some size dependency would be to obtain from the F07 PSDs the ratio between the third and second moments. This would then express the bulk optical properties as functions of the mean mass-weighted size of the PSD. However, here the intention is to relate directly the climate model prognostic second moment [i.e., IWC, because $m(D) = \text{constant} \times D^2$ to the bulk optical properties without the need to use diagnosed variables. The methodology adopted here, as previously mentioned, is equivalent to that of McFarquhar et al. (2003); in that paper, the effective radius of the PSD was expressed explicitly as a function of IWC, and then the bulk optical properties were themselves expressed as a function of the effective radius. This is equivalent to expressing the bulk optical properties as a function of IWC, which is the intention here.

Given β_{ext} and β_{sca} , then ω_0 can be found from

$$\omega_0 = \frac{\beta_{\text{sca}}}{\beta_{\text{ext}}},\tag{2}$$

and the bulk asymmetry parameter g is defined as follows:

$$g = \frac{\int g(\bar{l})C_{\rm sca}(\bar{l})n(\bar{l})\,d\bar{l}}{\int C_{\rm sca}(\bar{l})n(\bar{l})\,d\bar{l}},\tag{3}$$

where all the terms have been previously defined.

Equations (1)–(3) were used to compute the ensemble model bulk optical properties at 145 wavelengths between 0.2 and $120 \,\mu\text{m}$. To generate the F07 PSDs, the ensemble model mass-dimensional relationship was assumed. The PSDs themselves were estimated from a database consisting of 20 662 estimates of IWC and incloud temperature measurements obtained from a number of cirrus field campaigns, which were located in different regions of the world. The construction of the 20662 PSD database is fully described by Baran et al. (2014a). The latest estimates by Korolev et al. (2013) suggest that the impact of shattering on in situ derivations of IWC using historical PSDs could increase those estimates by up to about 30% at the heights and temperatures they considered. This does mean that some of the IWCs used here to generate the F07 PSDs could be similarly biased. However, the previously discussed remote sensing studies using the ensemble model optical properties at wavelengths relevant to a climate model indicate that the forward modeled measurements are generally within the uncertainties of those active and passive measurements. The wavelength resolution at which the ensemble model bulk optical properties were calculated is shown in Figs. 3a,b. The figures show the real and imaginary indices of solid ice as compiled by Warren and Brandt (2008), and the circles superimposed on the refractive indices are the wavelengths at which the bulk optical properties were calculated. It can be seen from Figs. 3a,b that the wavelength resolution used in this paper is sufficient to capture the rapid changes in the ice refractive index, especially between 1.5 and $4.0 \,\mu\text{m}$, and at wavelengths in the terrestrial window and far-infrared regions.

In the Met Office Unified Model, clouds are represented by vertical profiles of their mixing ratios with respect to air. Therefore, the volume extinction and scattering coefficients become the mass extinction and mass scattering coefficients. The spectral variation of the mass scattering coefficient K_{sca} and the asymmetry parameter g are shown as a function of q_i (i.e., the ice



FIG. 3. The (a) real and (b) imaginary refractive indices of ice compiled by Warren and Brandt (2008) where the marked full line represents the experimental values and the circles represent the 145 wavelengths at which the single-scattering properties were calculated.

mass mixing ratio) in Fig. 4a and Fig. 4b, respectively, where, in Fig. 4a, K_{sca} is a product of ω_0 and K_{ext} . The figure shows the variation of K_{sca} and g over the range of IWC used in the parameterization; this range goes from about 0.004 kg kg⁻¹ to about 9.0×10^{-9} kg kg⁻¹. These seven values of IWC were chosen from the possible 20662 values to qualitatively illustrate the dependence of K_{sca} and g on the shape of the PSD as a function of q_i . Figure 4a shows the spectral variation of K_{sca} , and it can be seen from the figure that at the highest values of q_i , K_{sca} is only weakly dependent on wavelength; this is because, at the highest values of q_i , the PSDs are very broad. Conversely, at the lowest values of q_i , the spectral variation of K_{sca} is significant, because the PSDs at those values are narrow. The figure also shows the wavelengths at which absorption is highest, such as between about 2.0 and 3.5 μ m and at 10.0 μ m and then again at around 40 μ m. Note, interestingly, in the far-infrared, K_{sca} increases between 20 and 30 μ m because of the imaginary index of solid ice decreasing



FIG. 4. The spectral dependence of (a) K_{sca} and (b) g as a function of the decadal logarithm of the IWC and wavelength. The values of IWC are shown by the color bar on the right-hand side of the figures.

between those wavelengths. Figure 4b shows the spectral variation of g as a function of wavelength and IWC. At the lowest wavelength of $0.20 \,\mu$ m, the asymmetry parameter has a value close to 0.70, and at the wavelength of $100 \,\mu$ m, g can decrease to values less than about 0.2. This behavior is expected at the longer wavelengths, as the ratio between ice crystal size and wavelength is small, and the IWCs are also very low, meaning very narrow PSDs. At around $3.0 \,\mu$ m, the asymmetry parameter can increase

to near 1.0; this is due to the very high ice absorption that occurs at that wavelength. To be computationally efficient in a climate or numerical weather prediction model, the simplest parameterizations of K_{ext} , ω_0 , and g are sought, but as can be seen from Fig. 4a and Fig. 4b, the variation of the bulk optical properties may be difficult to capture by simple parameterizations. The parameterization of the bulk optical properties is discussed and its accuracy tested in the section that follows.

TABLE 1. The values of the fitted prefactors and exponents estimated for each of the six Edwards and Slingo (1996) shortwave bands. The parameterization is from experiment 1, and it is equivalent to experiment 4.

λ	$a(\lambda)$	b(λ)	$c(\lambda)$	$d(\lambda)$
2.0×10^{-7} - 3.2×10^{-7}	51.5352	51.5352	0.783 862 87	0.003 550 69
3.2×10^{-7} - 6.9×10^{-7}	51.0986	51.0980	0.812 399 42	0.003 580 12
3.2×10^{-7} - 6.9×10^{-7}	51.0986	51.0980	0.812 399 4	0.003 580 12
6.9×10^{-7} -1.2×10^{-6}	51.5319	51.4624	0.821 957 75	0.003 780 12
1.2×10^{-6} - 2.4×10^{-6}	51.5569	44.6638	0.904 672 57	0.007 925 71
$2.4 \times 10^{-6} - 1.0 \times 10^{-5}$	51.5790	29.5045	0.992 610 77	0.007 321 16

4. The parameterization of the bulk optical properties

To first order, the mass extinction and mass scattering coefficients are the most important terms to be parameterized accurately to simulate the radiative transfer of solar and infrared radiation through cirrus. It can be seen from Fig. 4a that K_{sca} is only weakly dependent on wavelength, at the highest values of q_i , and that it only becomes strongly dependent on wavelength when $q_i <$ 10^{-4} kg kg⁻¹. This dependence of the first-order terms on wavelength suggests that a simple linear parameterization of these terms could suffice. As previously mentioned, the database consists of 20662 values of q_i , and for each of these values, there are 145 bulk optical properties. The total number of points in the database is, therefore, 2.996×10^6 . From this database, a total number of 28 values of q_i were selected to capture the monotonically decreasing values of q_i in near-regular spacing. The values of the q_i 28 points ranged from 0.004 to 10^{-9} kg kg⁻¹. With monotonically decreasing values of q_i , simple fits were found for K_{ext} , ω_0 , and g as a function of wavelength and q_i . The parameterized bulk optical properties are described by the following equations:

$$K_{\text{ext}}(\lambda, q_i) = a(\lambda)q_i, \qquad (4)$$

$$K_{\rm sca}(\lambda, q_i) = b(\lambda)q_i, \qquad (5)$$

$$\omega_0(\lambda, q_i) = K_{\rm sca}(\lambda, q_i) / K_{\rm ext}(\lambda, q_i), \quad \text{and} \qquad (6)$$

$$g(\lambda, q_i) = c(\lambda)q_i^{d(\lambda)}, \tag{7}$$

where the form of the above equations was estimated by trial and error; the prefactors shown in Eqs. (4), (5), and (7) were found by nonlinear least squares fitting; and λ is the wavelength. The values estimated for each of the prefactors and exponents are listed in Table 1 and Table 2, which were generated assuming the massdimensional relationship derived by Baran et al. (2011b) for the six shortwave and nine longwave bands, respectively. In Table 1 and Table 2, the wavelengths in each of the bands were not regularly spaced, as can be seen in Fig. 3a and Fig. 3b, so the parameterizations were implicitly weighted toward the part of each band that had the most wavelengths. The bulk optical properties described by Eqs. (4)–(7) were not band-averaged by the top-of-atmosphere solar and terrestrial irradiances. This is because band averaging should not be necessary, as the model is well resolved in spectral space, and errors in the parameterization and GCM will be more significant than differences in the bulk optical properties between their nonaveraged and band-averaged values. However, this latter statement was tested by averaging the bulk optical properties over the solar and terrestrial irradiances using the methodology of Lindner and Li (2000). It was found that the averaging had a small effect on the

TABLE 2. The values of the fitted prefactors and exponents estimated for each of the nine Edwards and Slingo (1996) longwave bands. The parameterization is from experiment 1 and it is equivalent to experiment 4.

λ	$a(\lambda)$	$b(\lambda)$	$c(\lambda)$	$d(\lambda)$
2.50×10^{-5} -1 $\times 10^{-02}$	52.4832	28.0152	1.341 704 44	0.059 153 16
1.82×10^{-5} - 2.5×10^{-5}	51.7668	30.0178	1.056 482 07	0.020 312 59
1.25×10^{-5} - 1.82×10^{-5}	51.6096	28.4440	1.01567484	0.01277607
1.33×10^{-5} - 1.70×10^{-5}	51.6145	28.4562	1.01870991	0.013 174 23
8.33×10^{-6} - 1.25×10^{-5}	51.5670	27.6883	1.010 484 76	0.007 129 82
8.92×10^{-6} - 1.01×10^{-5}	51.6293	28.0413	1.019 029 36	0.007 131 16
7.51×10^{-6} - 8.33×10^{-6}	51.6089	27.9997	1.02570536	0.009 016 60
6.66×10^{-6} -7.52 × 10 ⁻⁶	51.6039	27.6898	1.025 275 16	0.008 595 93
3.34×10^{-6} - 6.67×10^{-6}	51.5816	28.8421	1.026 916 49	0.010 811 90

prefactors and exponents presented in Tables 1 and 2 (not shown here for reasons of brevity), so its omission will not change any of the conclusions reached in this paper. The band limits shown in the tables are currently assumed in the GA configurations. It should be noted here that the parameterizations shown by the above equations are not fixed to the bands listed in the tables because they are functions of wavelength and can be applied to any of the GA configurations or any other climate model.

The ensemble model mass-dimensional relationship has been used here because it is consistent with the lidar and radiometric observations of semitransparent cirrus presented in Baran et al. (2014a). However, it is yet to be confirmed whether the mass-dimensional relationship used in Baran et al. (2014a) is applicable over a greater range of optical depth. The new bulk optical property parameterization is now compared against the E07 parameterization for three values of the IWC or q_i and two values of the environmental temperature. The three IWC and two temperature values assumed are 1.0×10^{-3} , 1.0×10^{-5} , and 1.0×10^{-7} kg kg⁻¹, and -60° and -30° C, respectively. As a reminder, the E07 parameterization assumes a relationship between D_e and the environmental temperature, and D_e is itself inversely related to the mass extinction coefficient. Therefore, at the coldest temperature, the mass extinction coefficient will be about twice as great as the mass extinction coefficient at the warmest temperature. These two temperatures will provide a range of mass extinction coefficients over which the new parameterization can be directly compared against that of E07. The range chosen for the ice mass mixing ratio is typical of what might be found in a climate model. The two bulk optical property parameterizations are compared against each other between the wavelength intervals of 2.0×10^{-7} to 3.2×10^{-7} m and 8.33×10^{-6} to 1.25×10^{-5} m, which are shortwave band 1 and longwave band 5 in the GCM. These two bands are chosen because shortwave spectral band 1 is largely nonabsorbing and longwave spectral band 5 covers the terrestrial window region as shown by Tables 1 and 2. The results of the comparisons are shown in Figs. 5a-c, where Fig. 5a shows the results for K_{ext} for shortwave spectral band 1, while Figs. 5b,c show the results for K_{abs} (which is simply K_{ext} – $K_{\rm sca}$) and the asymmetry parameter for longwave spectral band 5, respectively. In the figures, the results obtained at the cold and warm temperatures using the E07 parameterization are shown by the open and filled circles, respectively, and the new parameterization is shown by the diagonal cross symbol. Figure 5a shows that the E07 parameterization-predicted mass extinction coefficient will, at the coldest and warmest temperatures, be greater than Eq. (4) by factors of almost 2 and just over 1,

FIG. 5. A comparison between the E07 parameterization and the new bulk ice optical parameterization at shortwave spectral band 1 $(2.0 \times 10^{-7}-3.2 \times 10^{-7} \text{ m})$ and longwave spectral band 5 $(8.33 \times 10^{-6}-1.25 \times 10^{-5} \text{ m})$ as a function of IWC. The comparisons are for (a) K_{ext} at shortwave spectral band 1, (b) K_{abs} at longwave spectral band 5, and (c) the asymmetry parameter g at longwave spectral band 5. Keys are shown in the upper right-hand sides. The E07 parameterization is represented by the open and filled circles at the temperature values of -60° and -30° C, respectively. The new parameterization is represented by the diagonal cross symbol.



respectively. At shortwave spectral band 1, the singlescattering albedo is very close to 1 for both parameterizations, and the g value calculated using the E07 parameterization does not vary and is fixed, as a function of IWC, at a value of about 0.74. However, Eq. (7) varies between 0.77 and 0.74, between the highest and lowest values of IWC, respectively. This is because, at the highest value of IWC, the PSD is broader and will contain a greater occurrence of larger ice crystals than the smallest assumed value of IWC. This change in the shape of the PSD will naturally lead to larger and smaller g values, respectively. Because of the E07 parameterization being based on an invariant ice crystal model with respect to ice crystal size, at nonabsorbing wavelengths, its predicted g values cannot change irrespective of assumed temperature and/or IWC values.

Figure 5b shows that for the spectral band in the terrestrial window region, the E07 parameterization is about 50% more absorbing than the new parameterization at the lowest temperature at all values of IWC considered. However, at the temperature of -30° C, the new parameterization is about 10% more absorbing than the E07 parameterization. The change in ω_0 between the two temperatures using the E07 parameterization is largely invariant with IWC, changing from 0.50 to about 0.52, respectively, while Eq. (6) remains invariant with IWC and predicts a value of about 0.54 for all values of IWC considered. Figure 5c shows the expected invariance in the asymmetry parameter calculated using the E07 parameterization. However, the asymmetry parameter calculated using Eq. (7) decreases by about 6% between the highest and lowest values of IWC. This change in the asymmetry parameter reflects the change in the shape of the PSD, as previously discussed. The comparisons between the two parameterizations tend to show, at least at cirrus forming temperatures, that the E07 parameterization, for the same ice mass at the colder temperatures, should increase and decrease the outgoing SW and OLR relative to the new parameterization, respectively.

The accuracy of each of the parameterizations shown in Eqs. (4)–(7) is now examined using the full database of 2.995 \times 10⁶ values, minus the 28 values that have already been used to obtain the parameterizations. The error is calculated as the relative percent error, defined by the following equation:

$$\varepsilon(\lambda) = \frac{\operatorname{true}_{\lambda} - \operatorname{estimate}_{\lambda}}{\operatorname{true}_{\lambda}} \times 100\%, \qquad (8)$$

where the parameters true_{λ} and estimate_{λ} are the exact bulk optical properties obtained at each wavelength from the 20662 PSD database, which consists of 2.995 × 10⁶ values and their estimated values using Eqs. (4)–(7), respectively. The dependence of $\varepsilon(\lambda)$ on q_i has been



FIG. 6. The behavior of the relative percent error $\epsilon(\lambda)$ (shown by the key on the right-hand side of the figure) as a function of IWC (q_i) and wavelength derived from each of the following parameterized bulk optical properties: (a) K_{ext} , (b) ω_0 , and (c) g.

dropped from Eq. (8) for reasons of clarity. The results found for $\varepsilon(\lambda)$ are presented as a function of λ and q_i . The distribution of $\varepsilon(\lambda)$ in q_i and λ space is shown in Figs. 6a–c, respectively. The figures reflect, to some extent, the change in the imaginary refractive index of ice. At wavelengths in the near-infrared and window regions, at around 2.0–3.0 μ m, and at wavelengths greater than about 30 μ m, the $\varepsilon(\lambda)$ errors in K_{ext} and ω_0 are greatest, which is where solid ice is most absorbing. However, the errors are generally greatest for $q_i < 10^{-6} \text{ kg kg}^{-1}$. In general, in situ measurements of IWC down to such values are not, at the moment, possible, and current global model predictions of q_i are only occasionally less than $10^{-6} \text{ kg kg}^{-1}$. Moreover, the firstorder term is K_{ext} , which is observationally not known to within a conservative $\pm 50\%$ (Baran et al. 2009).

The relative errors in the parameterization for the asymmetry parameter at the longer wavelengths in the far-infrared generally follow Figs. 6a and 6b. However, in the important shortwave region and terrestrial window regions, the errors are within a few percent for all q_i values considered. However, for $\lambda \ge 40 \,\mu\text{m}$, the errors in the g parameterization generally increase, with the largest errors occurring when $q_i < 10^{-6} \text{ kg kg}^{-1}$. The relative errors found for g in the shortwave will not significantly bias the global model results. For all parameterizations of the bulk optical properties, the greatest density of the largest errors occurs in the farinfrared regions, where the infrared emission will have the lowest values and so will not contribute significantly to the OLR flux from the cloud, and in any case for $q_i <$ 10^{-6} kg kg⁻¹, the cloud optical depth will be very small at those wavelengths and q_i values.

The normalized probability distribution functions (PDFs) of $\varepsilon(\lambda)$, calculated for each of the three bulk optical properties, are presented in Figs. 7a–c. The mean of the PDF of $\varepsilon(\lambda)$ in K_{ext} shown in Fig. 6a is -2.96%, but the standard deviation is $\pm 14.7\%$. However, as previously stated, the experimental uncertainty in K_{ext} is about $\pm 50\%$, and since K_{ext} is the first-order term for radiative transfer, then the σ value in the K_{ext} relative error is well within the experimental uncertainty. Moreover, the parameterization is within $\pm 7.5\%$ for about 92% of all the cases considered.

The normalized PDF of $\varepsilon(\lambda)$ found for ω_0 is shown in Fig. 7b, and, as can be seen from the figure, the error is broader than Fig. 7a (not surprisingly, since the more difficult ice absorption regions are included in the parameterization). The mean of the ω_0 PDF is 14.02% with a σ value of 14.7%, with 74% of the cases being within $\pm 12.5\%$. The bias in the relative error calculated for ω_0 is positive, with a value of 0.031. A positive bias, with respect to the true value of ω_0 , could nudge the cloud to greater absorption (because of the predicted value of ω_0 being lower than the true value) and, therefore, warming, which could affect the temperature structure of the troposphere. However, this effect will be smaller than the impact of the first-order term K_{ext} .

The distribution of the normalized PDF found for g is shown in Fig. 7c, and the figure shows that the parameterized g is within $\pm 2.5\%$ of the true value of g for about 80% of all values. Moreover, there is no significant bias and the mean of the PDF is -2.133% with a σ value of $\pm 16.8\%$. However, this spread in g is primarily due to the longer wavelengths in the far-infrared, as shown in Fig. 6c,



FIG. 7. The normalized PDFs of $\epsilon(\lambda)$ derived from each of the following parameterized bulk optical properties: (a) K_{ext} (y axis runs from 0 to 0.6 and the x from -47.5% to +47.5%), (b) ω_0 (y axis runs from 0 to 0.25 and the x from -77.5% to +47.5%), and (c) g (y axis runs from 0 to 0.45 and the x from -47.5% to +27.5%).



FIG. 8. The 20-yr area-averaged differences between the GA5 model and CERES measurements of the reflected SW and OLR at the TOA. Flux differences (W m⁻²) are shown by the color bars. The SW differences (a) assuming the GA5 control model and (b) experiment 1. The LW differences (c) assuming the GA5 control model and (d) experiment 1.

which in the PDF occur very infrequently. The impact of the parameterizations on the global model is discussed in the next section.

5. The impact of the parameterizations on GA5: A series of experiments

In this section, the impacts of the parameterizations on the GA5 configuration of the global model are presented in the form of a series of experiments under various assumptions. As previously stated in the introduction, each of the experiments is run over a 20-yr period, starting in 1989 and finishing in 2008. Each of the experiments is compared against the latest CERES annual 20-yr averaged shortwave and longwave mean fluxes at TOA, as compiled by Stephens et al. (2012), which uses the CERES reanalysis performed by Loeb et al. (2009). Each of the experiments is also compared against the ERA-Interim analysis of atmospheric temperature observations covering the same 20-yr period (Dee et al. 2011). The control run assumes the following ice microphysics and ice optical parameterization. The massdimensional relationship is assumed to be the one from Brown and Francis (1995), and the PSDs are derived from the Houze et al. (1979) parameterization. The ice optical parameterization used in the control run is taken from E07. Each of the experiments is described in the next few paragraphs:

Experiment 1: Consistent PSD and inconsistent mass-dimensional relationship. The ice microphysics assumed in this experiment is taken from Furtado et al. (2014), and the F07 PSD parameterization is used for the ice crystal size spectrum. The ice optical parameterization is taken from Tables 1 and 2. In the radiation scheme, the F07 PSDs are generated using the mass-dimensional relationship predicted by the ensemble model (i.e., mass = $0.04D^2$). The bulk mass extinction and scattering coefficients are determined by applying different weights to each member of the ensemble model for each of the size bins in the PSD. The



FIG. 9. The differences in zonally averaged temperature (K) between the GA5 model and the ERA-Interim temperature product as a function of altitude (hPa) and latitude assuming (a) the control model and (b) experiment 1. The temperature differences are shown by the color bar.

weighted-averaged mass extinction and mass scattering coefficients $\langle K_{\text{ext/sca}} \rangle$ are therefore given by the following equation:

$$\langle K_{\text{ext/sca}} \rangle = \int_{D_{\min}}^{D_{\max}} \left[\sum_{j=1}^{6} w_j k_j(D) \right] n(D) \, dD \,, \qquad (9)$$

where D_{\min} and D_{\max} are the minimum and maximum ice crystal dimensions in the PSD, n(D) is the F07 PSDs, and w_i is the weight applied to the *j*th ensemble member, where by definition $\Sigma w_i = 1$ for each of the size bins in the PSD. In Eq. (9) the subscripts ext/sca have been removed from $k_i(D)$ for reasons of clarity. Obviously, $K_{\text{ext/sca}}$ are the mass extinction and mass scattering cross sections predicted for each ensemble member of maximum dimension D. The weights in this experiment are assumed to be 0.90, 0.07, and 0.03, applied to the first three ensemble model members, respectively. These weights are chosen because they are the same as those used by Baran et al. (2014a) in evaluating the ensemble model against solar and infrared measurements obtained above optically thin cirrus. However, it is as yet unclear if these weights are applicable over a broader range of optical depth than was sampled by Baran et al. (2014a).

Experiment 2: Consistent PSD and mass-dimensional relationship. In this experiment, the same ice microphysics is assumed as in experiment 1, except that in the radiation scheme, the F07 PSDs are generated assuming the Cotton et al. (2013) mass-dimensional relationship (i.e., mass = $0.0257D^2$), and these PSDs are used to

predict the bulk ice optical properties. The weights applied in this experiment to obtain the weighted-average mass extinction and scattering coefficients are the same as those assumed in experiment 1.

Experiment 3: Fully consistent but ensemble model reweighted toward more ice aggregated members. This experiment assumes, in the radiation scheme, the same mass-dimensional relationship as used in experiment 2, and the ice microphysics is kept the same as in experiment 1, except that in this experiment the ensemble model mass extinction is weighted toward the more aggregated members of the ensemble. The weights assumed here are 0.1, 0.2, 0.1, 0.4, 0.1, and 0.1, respectively.

Experiment 4: Fully consistent but ensemble model reweighted toward less aggregated members. In this experiment, the control run has been changed. Here the control ice microphysics is the same as in experiment 1, but the radiation scheme, which is the E07 parameterization, is kept the same as the previous control. In the experimental run, the F07 PSDs used in the radiation scheme have been generated assuming the same massdimensional relationship as used in experiment 2, but the ensemble model is weighted so that the mass extinction coefficients are equivalent to the mass extinction coefficients of experiment 1. To obtain this equivalence, the ensemble model was weighted using the following set of weights: 0.5, 0.2, 0.3, 0, 0, and 0. Here, the PSDs and mass-dimensional relationships assumed in the cloud and radiation schemes of GA5 are completely consistent. This experiment is hereafter referred to as the "consistent radiation treatment," as this



FIG. 10. As in Fig. 8, but for experiment 2, showing (a) SW flux differences at TOA between experiment 2 and CERES and (b) OLR flux differences at TOA. In this experiment, the PSDs are generated assuming mass = $0.0257D^2$ and the weighted ensemble model is the same as in experiment 1.

experiment is consistent with the solar and radiometric observations of Baran et al. (2014a) but using the Cotton et al. (2013) derived mass-dimensional relationship to generate the PSDs. The results obtained from each of the experiments are now discussed in sequential order.

a. Experiment 1

As a reminder, in this experiment, the new ice microphysics is utilized, and the ensemble model massdimension relationship is used to generate the F07 PSDs. The shortwave and longwave cirrus radiative properties are calculated using the new ice optical parameterization given in Tables 1 and 2, respectively. The shortwave and longwave flux differences at TOA between the simulations and the CERES observations are shown in Figs. 8a–d. As shown in Fig. 8b, experiment 1 leads to a slight improvement in the area-averaged SW root-mean-square error (rmse) of 0.13 Wm^{-2} . However, there are discernible divergences in the spatial distribution of the differences between the control and



experiment 1. For instance, in the tropics, experiment 1 is not as bright as the control, especially off the coast around Brazil and off the West African coast near the equator, as well as centered on the equator over the Indian Ocean. Note also, experiment 1 over Antarctica is significantly less bright than the control. There is also a distinct decrease in the shortwave reflection, relative to the control, around the west Atlantic Ocean to near-neutral values and around the UK. The OLR differences are shown in Figs. 8c and 8d.

In the longwave, the new parameterization improves the area-averaged rmse by 0.21 W m^{-2} . Moreover, off the coast of Antarctica, the new parameterization decreases the OLR differences to generally neutral values, although in places, such as the high midlatitudes over the Atlantic Ocean and over Canada, experiment 1 decreases the cloud transmission relative to the control. In general, over the tropics, experiment 1 increases the OLR relative to the control by about $5-10 \,\mathrm{Wm^{-2}}$; therefore, the ice cloud in experiment 1 is generally more transmitting in that region relative to the control. With respect to the control, the new parameterization does not have any undesirable impacts on the radiative properties of the model. However, the new parameterization has its biggest impact on the model in terms of the temperature structure of the troposphere, as shown in Fig. 9b. The figure shows that, in the tropical troposphere, experiment 1 removes the cold bias that is evident in the control shown in Fig. 9a, probably because the ice cloud in the control, over the western Pacific, is more transmitting relative to the cloud in experiment 1. However, experiment 1 does produce a slight warm bias of about 0.5 K in the southern midlatitudes. In general, in the climatically important tropical region, experiment 1 and the new ice optical parameterization clearly have a positive impact. The impact of changing the massdiameter relationship in the radiation scheme is discussed in the next experiment.

b. Experiment 2

In this experiment, the ice microphysics is the same as experiment 1, but the F07 PSDs in the radiation scheme are generated using the Cotton et al. (2013) massdimensional relationship, and these PSDs are used to predict the ice bulk optical properties. The impact of experiment 2 on the SW and LW flux differences at TOA is shown in Figs. 10a and 10b, respectively. The impact of experiment 2 on the temperature structure of the troposphere is shown in Fig. 11. Clearly, the figures show that, on broadening the PSD in the radiation scheme while keeping the size-shape distribution the same as in experiment 1, the impact on the radiative fluxes and temperature structure of the troposphere is clearly considerable. In this experiment, relative to experiment 1, the area-weighted SW and LW rmse have increased by 0.68 and 1.29 W m⁻², respectively. Both Figs. 10a and 10b show that, on broadening the PSD, there is considerably more SW flux reflected back to space and considerably less OLR relative to experiment 1. In general, it can be concluded from experiment 2 that the ice cloud is generally less transmitting than the ice cloud predicted in experiment 1. The impact of less transmitting cloud on the temperature structure of the troposphere is demonstrated in Fig. 11. The figure shows that the broadening of the PSD, which allows the occurrence of larger ice crystals relative to those in experiment 1, results in more absorption and therefore more warming of the troposphere at high altitudes in the tropics by about 0.5 K and by about 1 K in the southern midlatitudes. There is also some warming, of about 0.5 K, at altitudes greater than 600 hPa at northern midlatitudes.

Clearly, this experiment demonstrates the need to constrain the general shape of the PSD and determine more accurate mass-dimensional relationships. The impact of changing the ensemble shape distribution while preserving the shape of the PSD to be the same as that used in experiment 2 is discussed in the next experiment.

c. Experiment 3

For this experiment, the same ice microphysics and PSDs are used as in the previous experiment, but the ensemble model is reweighted toward the more ice aggregated members to increase ice cloud transmission. The results of experiment 3 for the SW and LW and its impact on the temperature structure of the troposphere are shown in Fig. 12a, Fig. 12b, and Fig. 13, respectively. Relative to experiment 2, increasing the ice cloud transmission by reweighting the ensemble model to more aggregated members results in less bright cloud



FIG. 12. As in Fig. 10, but for experiment 3. In this experiment, the PSDs are generated assuming mass $= 0.0257D^2$ and the ensemble model is weighted toward the more ice aggregated members.

and greater transmission in the OLR, as shown in Figs. 12a and 12b, respectively. Some of the SW impacts are desirable, especially over the Atlantic Ocean, where differences between the model and CERES are close to neutral values. This is also true in certain parts of the Southern Ocean but is by no means general. Interestingly, over Antarctica the impact of experiment 3 is similar to that of experiment 1, and both lead to an improvement over that region relative to the control. However, in general, experiment 3 clearly increases the

area-weighted rmse by 1.7 W m^{-2} relative to experiment 1, which is undesirable.

The impact of increasing the cloud transmission, relative to experiment 1, is to reintroduce the cold bias seen in the control in the tropical troposphere, as shown in Fig. 13. It is also clear that experiment 3 leads to an increase in the cold bias in the lower atmosphere at high northern midlatitudes relative to the control and experiment 1. This experiment demonstrates that changing the values of the weights applied to habit mixture





models of cirrus will also have an important impact on the radiative fluxes at TOA and on the temperature structure of the atmosphere. Therefore, it is of necessity to constrain not only the shape of the PSD but also the size-shape distribution across the PSD. Experiments 2 and 3 demonstrate that the shape of the PSD and the weighting of habit mixture models are of equal importance.

d. Experiment 4

In this last experiment, the F07 PSDs are the same as those used in experiment 2, but the ensemble model is reweighted so that the ice mass extinction and scattering coefficients are the same as those used in experiment 1. In the case of the control run, the ice microphysics is the same as was used in experiment 1, but the ice cloud radiation scheme is kept the same as in the previous control run. Figure 14 shows the reflected shortwave (Figs. 14a,b) and OLR (Figs. 14c,d) differences at TOA between E07 and the new ice optical parameterizations and the CERES observations. The figure shows that the consistent radiation treatment produces less bright cloud and more longwave cloud transmission relative to the E07 parameterization, which is an expected result from Figs. 5a and 5b. However, the bias toward more longwave cloud transmission results in more areas of near-zero differences between model and observation relative to the E07 parameterization. This is especially true in the Southern Ocean, and areas in the midlatitudes, as shown in Fig. 14d. The results obtained from this consistent radiation treatment are very similar to the results from the inconsistent experiment 1 (Figs. 8b,d with SW rmse = 8.61 W m^{-2} and LW rmse = 6.97 W m^{-2}).

To understand these differences between the two radiation schemes and the CERES observations, Figs. 15a and 15b show the SW and LW flux differences between the consistent radiation and the E07 parameterization. Figure 15a shows that most of the SW differences between the two radiation schemes occur in the tropics, where the consistent radiation treatment predicts cloud that is generally less bright than the E07 parameterization by about 10 W m^{-2} , and in the tropical warm pool region this difference can be up to about 15 W m^{-2} . The outgoing LW differences between the two schemes are shown in Fig. 15b, and from this figure, the consistent radiation treatment allows more cloud LW transmission than the E07 parameterization. Outside of the tropics, the differences are up to about 5 W m^{-2} , while in the tropics the differences can be up to about 15 W m^{-2} . The higher LW differences are generally around the tropical warm pool.

Given these SW and LW differences between the two radiation schemes and the fact that the consistent radiation is a coupling between the microphysics and bulk optical properties, it would be interesting to compare differences in the SW and LW radiative effect between the two schemes and the CERES observations. The SW and LW radiative effect differences between the model and CERES are shown in Figs. 16a-d, assuming the E07 parameterization and the consistent radiation scheme, respectively. Figure 16a demonstrates that, given improved microphysics, the SW radiative effect that the E07 parameterization predicts is sometimes overcooling relative to the CERES product. In the tropics and midlatitudes, these differences can be up to about $-15 \,\mathrm{W}\,\mathrm{m}^{-2}$, and in some tropical locations the radiative effect differences can reach up to about $-30 \,\mathrm{W \,m^{-2}}$ off the coast of South America and along the equator above the Indian Ocean. The SW radiative effect predicted by the consistent scheme is shown in Fig. 16b, and this essentially shows the opposite to the E07 parameterization. The consistent scheme generally decreases the cooling predicted by the E07 parameterization, although relative to CERES, this can appear as too much warming by up to about $30 \,\mathrm{Wm^{-2}}$ in some locations in the tropics and especially around the tropical warm pool. However, in the midlatitude regions (with the notable exception of the tropical warm pool), the consistent radiation scheme appears to decrease the cooling predicted by the E07 parameterization, which appears desirable relative to the CERES product.

The LW radiative effect predicted by the E07 parameterization and the consistent radiation scheme is shown in Figs. 16c and 16d, respectively. The two schemes, relative to the CERES product, generally appear more similar to each other. However, in the tropics and midlatitudes, the consistent radiation scheme tends to offset the E07 LW radiative effect more in the direction of the CERES observations. This similarity in the longwave radiative effect between the two schemes



FIG. 14. As in Fig. 8, but for experiment 4. In this experiment, the control uses the same ice microphysics as used in experiment 1 and the ensemble model is reweighted such that the ice mass extinction and scattering coefficients are equivalent to experiment 1. The SW differences assuming (a) the E07 ice optics parameterization and (b) the consistent radiation treatment. The LW differences assuming (c) the E07 ice optics parameterization and (d) the consistent radiation treatment. In both the (left) control and (right) experiments, the same microphysics is assumed in the cloud scheme.

could be because the model predicts the cloud to be at certain altitudes, and it is the radiating temperature of these clouds at those altitudes that dominates the CERES measurements, rather than differences in the radiation parameterizations. The differences in the SW and LW radiative effects between the consistent radiation scheme and the E07 parameterization are shown in Figs. 17a and 17b, respectively. Figure 17a shows that the consistent radiation scheme generally predicts less SW cooling than the E07 parameterization by about 5 W m⁻² outside of the tropics; in the tropics this can reach up to about 15 W m⁻², while over the tropical warm pool, the difference can reach up to about 20 W m⁻². Figure 17b shows that the differences in the LW radiative effect

between the two radiation schemes occurs either in the tropics, generally around the tropical warm pool, or at high latitudes in both hemispheres, and the signs are opposite. Not surprisingly, given previous results, over the tropical warm pool the consistent radiation predicts less LW radiative effect than the E07 parameterization by about -10 Wm^{-2} , while at the high latitudes in both hemispheres, the consistent radiation scheme predicts more of an LW radiative effect than the E07 parameterization, by up to about 10 Wm^{-2} .

The results presented in this section have shown that, given any mass-dimensional relationship, the equivalent fluxes, or for that matter, radiances can be predicted by reweighting the ensemble model, given observational



FIG. 15. Differences between the consistent radiation scheme and the E07 parameterization in the (a) reflected SW flux at TOA and (b) the emergent LW flux at TOA.

radiometric constraints on the distribution of weights or in situ estimates of the size-shape PSD.

The impacts of the E07 and consistent radiation parameterizations on the temperature structure of the troposphere are shown in Figs. 18a and 18b, respectively. Figure 18b shows that the consistent radiation treatment is generally similar to experiment 1 (Fig. 9b), most notably, with a further slight reduction of the cold bias in the tropical troposphere relative to Fig. 18a. This same cold bias is also significantly reduced relative to the original control by the E07 parameterization when convolved

with the more representative cloud microphysics derived from Furtado et al. (2014) (Fig. 18a). There is a further slight warming of southern midlatitudes in the troposphere produced by the consistent radiation treatment relative to the E07 parameterization. Differences in the zonal mean temperature structure of the atmosphere between the consistent radiation scheme and the E07 parameterization are shown in Fig. 19, and this figure shows that the consistent radiation scheme is generally more warming than the E07 parameterization by up to about 1 K.



FIG. 16. As in Fig. 14, but for the SW and LW radiative effect differences between the model and the CERES product assuming the E07 parameterization and a consistent radiation scheme, respectively. (a) SW radiative effect differences between the E07 parameterization and the CERES product. (b) As in (a), but for the consistent radiation scheme. (c) LW radiative effect differences between the E07 parameterization and CERES product. (d) As in (c), but for the consistent radiation scheme.

This experiment has shown that the consistent radiation treatment predicts cloud that is generally less bright and increases the OLR relative to the E07 parameterization, and generally warms the zonal mean temperature structure of the atmosphere by up to about 1 K. These effects have the consequence that the consistent radiation scheme will predict less of an SW radiative effect than the E07 parameterization. However, differences in the LW radiative effect between the consistent radiation scheme and the E07 parameterization mostly occur around the tropical warm pool, where they are negative, and at high latitudes in both hemispheres, where the differences are positive. Therefore, given the same microphysics scheme in GA5, the consistent radiation scheme predicts less of an LW radiative effect around the tropical warm pool but more of an effect at high latitudes in both hemispheres. A comparison between the CERES global means predicted by the consistent radiation treatment and the E07 parameterization is shown in Table 3. The Table shows that the E07 parameterization has increased the outgoing shortwave and decreased the outgoing longwave fluxes at

TOA relative to the consistent radiation treatment, in accord with expectation. This is most notable in the shortwave, where the E07 parameterization now exceeds the observational uncertainty in the outgoing shortwave flux at TOA. It is expected that given further improvements to the global model, which may result in more ice mass being predicted, the E07 parameterization will result in greater divergences between model and observation. However, Table 3 also shows that the consistent radiation treatment in the absorbed shortwave and SW CRE are both outside the observational uncertainty. It is expected that, with an improved climate model prediction of ice mass, these two SW radiative components should agree within the observational uncertainty. A further possibility as to why the consistent radiation treatment underpredicts some of the SW components shown in Table 3 could also be due to the asymmetry parameter predicted by the ensemble model not being sufficiently low. However, for now, it is more likely that improving the representation of ice mass in the Met Office GA configuration will improve the consistent



FIG. 17. As in Fig. 15, but for (a) differences in the SW radiative effect and (b) differences in the LW radiative effect between the consistent radiation scheme and the E07 parameterization.

radiation prediction of those two SW components. Although the E07 parameterization is not detrimental to the performance of GA5 with improved microphysics, with further improvements to the model, the E07 parameterization can only get worse when compared against observation. This is because the cancellation of error that was discussed in the introduction will be removed through further model improvements. Consistent physics within predictive models must always be preferred over inconsistent treatments. Unfortunately, the inconsistent treatments are currently prevalent in climate models, which may give reasonable predictions, but for the wrong physical reasons.

6. Conclusions and discussion

A new flexible shortwave and longwave parameterization of the ensemble model bulk optical properties has been presented and applied to the Met Office Unified Model Global Atmosphere 5.0 configuration (GA5). The bulk optical properties have been computed between the wavelengths of 0.2 and $120 \,\mu$ m at sufficient resolution in wavelength space such that the rapid variation of the imaginary refractive index of ice in the near-infrared and longwave parts of the spectrum is captured. It has been shown that the parameterization can reproduce the database of 2.99×10^6 bulk mass extinction coefficients to



FIG. 18. As in Fig. 9, but for experiment 4: (a) assuming the E07 ice optics parameterization and (b) assuming the consistent radiation treatment. Again, in both the (a) control and (b) experiment, the same microphysics is assumed in the cloud scheme.

within $\pm 7.5\%$ for over 90% of the database. The largest errors occur at wavelengths in the far-infrared. At these wavelengths, the Planck function will be small and so will not bias the values of mass extinction. Currently, the mass extinction of cirrus is not known experimentally to within $\pm 50\%$. The mass extinction is the first-order term that determines the reflection and transmission properties of cloud. The error in the mass extinction parameterization is well within the current experimental uncertainty. The single-scattering albedo and asymmetry parameter parameterizations were found to be within $\pm 12.5\%$ and $\pm 2.5\%$ for 74% and 80% of the database, respectively. The largest errors in the single-scattering albedo and asymmetry parameter tended to occur at the lowest values of IWC and occurred in the far-infrared regions.

The new parameterization is more flexible than previous parameterizations used in the Met Office series of GA models, since the weighting of the ensemble members can be changed, given some mass-dimensional relationship to generate the parameterized PSD, and this preserves physical consistency between the cloud physics and radiation schemes of the climate model. This flexibility



FIG. 19. Differences in the zonal mean temperature between the consistent radiation scheme and the E07 parameterization.

TABLE 3. CERES means obtained from Stephens et al. (2012) compared against the GA5 means predicted using the E07 parameterization and the consistent radiation treatment (units: $W m^{-2}$).

		E07	a
Variable	Observation	parameterization	Consistent
Outgoing SW	100 ± 2	102.23	99.01
Outgoing LW	239.7 ± 3.3	237.69	239.31
Absorbed SW	240.2 ± 2	239.16	242.38
Albedo TOA	29.41 ± 1	29.94	29.00
SW radiative effect	-47.5 ± 3	-47.09	-43.96
LW radiative effect	26.7 ± 4	25.88	24.62
Net surf SW	165 ± 6	167.56	167.14

was demonstrated in experiments 3 and 4. Experiments 2, 3, and 4 demonstrate that the weighting of habit mixture models and assumed shapes of the PSDs or mass–*D* relationships are of equal importance.

The paper has demonstrated that the overarching philosophy in demanding consistent cloud physics between the cloud and radiation schemes does not have detrimental effects on the overall performance of GA5. Rather, if this overarching philosophy is not followed, as demonstrated by the control used in experiment 1 and the application of the E07 parameterization used as the control in experiment 4, then inconsistent microphysics may lead to desirable results, but for perhaps the wrong physical reasons. Insisting on consistency between the cloud and radiation schemes means that all other parameterizations that are predicting the cloud macrophysical and microphysical properties must realistically represent the most pertinent processes relevant to those parameterizations.

The experiments presented in this paper demonstrate the importance of observationally constraining the shape of the PSD and which general mass-dimensional relationship to apply to climate models. Furthermore, just as important as this is to determine experimentally which size-shape distribution (in terms of a weighted habit mixture model of cirrus) best matches global SW and LW radiometric observations.

Relying on error cancellation, as discussed at the beginning of this paper and in the discussion of experiment 4, is undesirable, as when the distributions and amounts of IWC are finally addressed in climate models, those parameterizations that do rely on error cancellation can only get worse (assuming IWC is increased to the observed amounts and at the correct altitude). This, to some extent, has already been identified in the results presented in Table 3. A further overarching philosophy of the approach in this paper is to remove error cancellation by directly relating global model prognostic variables to ice optical properties. In this way, global model prognostic variables are directly tested against radiative measurements, rather than placing a reliance on diagnosed variables, which may lead to error cancellation; this will then preclude identification of systematic climate model biases.

Acknowledgments. We thank Professor Greg McFarquhar and two anonymous reviewers for their helpful comments.

REFERENCES

- Baran, A. J., 2005: The dependence of cirrus infrared radiative properties on ice crystal geometry and shape of the sizedistribution function. *Quart. J. Roy. Meteor. Soc.*, **131**, 1129– 1142, doi:10.1256/qj.04.91.
- —, 2007: The impact of cirrus microphysical and macrophysical properties on upwelling far-infrared spectra. *Quart. J. Roy. Meteor. Soc.*, **133**, 1425–1437, doi:10.1002/qj.132.
- —, 2009: A review of the light scattering properties of cirrus. J. Quant. Spectrosc. Radiat. Transf., 110, 1239–1260, doi:10.1016/ j.jqsrt.2009.02.026.
- —, 2012: From the single-scattering properties of ice crystals to climate prediction: A way forward. *Atmos. Res.*, **112**, 45–69, doi:10.1016/j.atmosres.2012.04.010.
- —, and P. N. Francis, 2004: On the radiative properties of cirrus cloud at solar and thermal wavelengths: A test of model consistency using high-resolution airborne radiance measurements. *Quart. J. Roy. Meteor. Soc.*, **130**, 763–778, doi:10.1256/qj.03.151.
- —, and L.-C. Labonnote, 2007: A self-consistent scattering model for cirrus. I: The solar region. *Quart. J. Roy. Meteor. Soc.*, **133**, 1899–1912, doi:10.1002/qj.164.
- —, P. J. Connolly, and C. Lee, 2009: Testing an ensemble model of cirrus ice crystals using midlatitude in situ estimates of ice water content, volume extinction coefficient, and the total solar optical depth. J. Quant. Spectrosc. Radiat. Transf., 110, 1579–1598, doi:10.1016/j.jqsrt.2009.02.021.
- —, A. Bodas-Salcedo, R. Cotton, and C. Lee, 2011a: Simulating the equivalent radar reflectivity of cirrus at 94 GHz using an ensemble model of cirrus ice crystals: A test of the Met Office global numerical weather prediction model. *Quart. J. Roy. Meteor. Soc.*, **137**, 1547–1560, doi:10.1002/qj.870.
- —, P. J. Connolly, A. J. Heymsfield, and A. Bansemer, 2011b: Using in situ estimates of ice water content, volume extinction coefficient, and the total solar optical depth obtained during the tropical ACTIVE campaign to test an ensemble model of cirrus ice crystals. *Quart. J. Roy. Meteor. Soc.*, 137, 199–218, doi:10.1002/qj.731.
- —, R. Cotton, K. Furtado, S. Havemann, L.-C. Labonnote, F. Marenco, A. J. Smith, and J.-C. Thelen, 2014a: A self-consistent scattering model for cirrus. II: The high and low frequencies. *Quart. J. Roy. Meteor. Soc.*, **140**, 1039–1057, doi:10.1002/qj.2193.
- —, K. Furtado, L.-C. Labonnote, S. Havemann, J.-C. Thelen, and F. Marenco, 2014b: On the relationship between the scattering phase function of cirrus and the atmospheric state. *Atmos. Chem. Phys. Discuss.*, **14**, 14109–14157, doi:10.5194/acpd-14-14109-2014.
- Baum, B. A., A. J. Heymsfield, P. Yang, and S. T. Bedka, 2005: Bulk scattering properties for the remote sensing of ice clouds. Part I: Microphysical data and models. J. Appl. Meteor., 44, 1885–1895, doi:10.1175/JAM2308.1.
- —, P. Yang, A. J. Heymsfield, C. G. Schmitt, Y. Xie, A. Bansemer, Y.-X. Hu, and Z. Zhang, 2011: Improvements in shortwave bulk

scattering and absorption models for the remote sensing of ice clouds. *J. Appl. Meteor. Climatol.*, **50**, 1037–1056, doi:10.1175/2010JAMC2608.1.

- Bozzo, A., T. Maestri, R. Rizzi, and E. Tosi, 2008: Parameterization of single scattering properties of mid-latitude cirrus clouds for fast radiative transfer models using particle mixtures. *Geophys. Res. Lett.*, **35**, L16809, doi:10.1029/2008GL034695.
- Brown, A. R., S. Milton, M. Cullen, B. Golding, J. Mitchell, and A. Shelly, 2012: Unified modeling and prediction of weather and climate: A 25-year journey. *Bull. Amer. Meteor. Soc.*, 93, 1865–1877, doi:10.1175/BAMS-D-12-00018.1.
- Brown, P. R. A., and P. N. Francis, 1995: Improved measurements of the ice water content in cirrus using a total-water probe. J. Atmos. Oceanic Technol., 12, 410–414, doi:10.1175/ 1520-0426(1995)012<0410:IMOTIW>2.0.CO;2.
- Cotton, R. J., and Coauthors, 2013: The effective density of small ice particles obtained from in situ aircraft observations of midlatitude cirrus. *Quart. J. Roy. Meteor. Soc.*, **139**, 1923–1934, doi:10.1002/qj.2058.
- Dee, D. P., and Coauthors, 2011: The ERA-Interim reanalysis: Configuration and performance of the data assimilation system. *Quart. J. Roy. Meteor. Soc.*, **137**, 553–597, doi:10.1002/gj.828.
- Delanoë, J., R. J. Hogan, R. M. Forbes, A. Bodas-Salcedo, and T. H. M. Stein, 2011: Evaluation of ice cloud representation in the ECMWF and UK Met Office models using CloudSat and CALIPSO data. *Quart. J. Roy. Meteor. Soc.*, **137**, 2064–2078, doi:10.1002/qj.882.
- De Leon, R. R., and J. D. Haigh, 2007: Infrared properties of cirrus clouds in climate models. *Quart. J. Roy. Meteor. Soc.*, 133, 273– 282, doi:10.1002/qj.1.
- Edwards, J. M., and A. Slingo, 1996: Studies with a flexible new radiation code. I: Choosing a configuration for a large-scale model. *Quart. J. Roy. Meteor. Soc.*, **122**, 689–719, doi:10.1002/ qj.49712253107.
- —, S. Havemann, J.-C. Thelen, and A. J. Baran, 2007: A new parameterization for the radiative properties of ice crystals: Comparison with existing schemes and impact in a GCM. *Atmos. Res.*, 83, 19–35, doi:10.1016/j.atmosres.2006.03.002.
- Field, P. R., and A. J. Heymsfield, 2003: Aggregation and scaling of ice crystal size distributions. J. Atmos. Sci., 60, 544–560, doi:10.1175/ 1520-0469(2003)060<0544:AASOIC>2.0.CO;2.
- —, R. Wood, P. R. A. Brown, P. H. Kaye, E. Hirst, R. Greenaway, and J. A. Smith, 2003: Ice particle interarrival times measured with a fast FSSP. J. Atmos. Oceanic Technol., 20, 249–261, doi:10.1175/1520-0426(2003)020<0249:IPITMW>2.0.CO;2.
- —, A. J. Heymsfield, and A. Bansemer, 2006: Shattering and particle interarrival times measured by optical array probes in ice clouds. J. Atmos. Oceanic Technol., 23, 1357–1371, doi:10.1175/ JTECH1922.1.
- —, —, and —, 2007: Snow size distribution parameterization for midlatitude and tropical ice cloud. J. Atmos. Sci., 64, 4346– 4365, doi:10.1175/2007JAS2344.1.
- —, —, , and C. H. Twohy, 2008: Determination of the combined ventilation factor and capacitance for ice crystal aggregates from airborne observations in a tropical anvil cloud. *J. Atmos. Sci.*, **65**, 376–391, doi:10.1175/2007JAS2391.1.
- —, R. J. Cotton, K. McBeath, A. P. Lock, S. Webster, and R. P. Allan, 2014: Improving a convection-permitting model simulation of a cold air outbreak. *Quart. J. Roy. Meteor. Soc.*, **140**, 124– 138, doi:10.1002/qj.2116.
- Francis, P. N., 1995: Some aircraft observations of the scattering properties of ice crystals. J. Atmos. Sci., 52, 1142–1154, doi:10.1175/ 1520-0469(1995)052<1142:SAOOTS>2.0.CO;2.

- —, A. Jones, R. W. Saunders, K. P. Shine, A. Slingo, and Z. Sun, 1994: An observational and theoretical study of the radiative properties of cirrus. Some results from ICE'89. *Quart. J. Roy. Meteor. Soc.*, **120**, 809–848, doi:10.1002/qj.49712051804.
- Fu, Q., 2007: A new parameterization of an asymmetry factor of cirrus clouds for climate models. J. Atmos. Sci., 64, 4140–4150, doi:10.1175/2007JAS2289.1.
- —, W. B. Sun, and P. Yang, 1999: Modeling of scattering and absorption by nonspherical cirrus ice particles at thermal infrared wavelengths. J. Atmos. Sci., 56, 2937–2947, doi:10.1175/ 1520-0469(1999)056<2937:MOSAAB>2.0.CO;2.
- Furtado, K., P. R. Field, R. Cotton, A. J. Baran, 2014: The effects of ice particle fall speed and size distribution on simulated high cloud. *Quart. J. Roy. Meteor. Soc.*, in press.
- Goody, R. M., J. G. Anderson, and G. R. North, 1998: Testing climate models: An approach. *Bull. Amer. Meteor. Soc.*, **79**, 2541–2549, doi:10.1175/1520-0477(1998)079<2541:TCMAA>2.0.CO;2.
- Gu, Y., K. N. Liou, S. C. Ou, and R. Fovell, 2011: Cirrus cloud simulations using WRF with improved radiation parameterization and increased vertical resolution. J. Geophys. Res., 116, D06119, doi:10.1029/2010JD014574.
- Heymsfield, A. J., and L. M. Miloshevich, 2003: Parameterizations for the cross-sectional area and extinction of cirrus and stratiform ice cloud particles. J. Atmos. Sci., 60, 936–956, doi:10.1175/ 1520-0469(2003)060<0936:PFTCSA>2.0.CO;2.
- —, C. Schmitt, A. Bansemer, and C. H. Twohy, 2010: Improved representation of ice particle masses based on observations in natural clouds. J. Atmos. Sci., 67, 3303–3318, doi:10.1175/ 2010JAS3507.1.
- Hong, G., P. Yang, B. A. Baum, A. J. Heymsfield, and K.-M. Xu, 2009: Parameterization of shortwave and longwave radiative properties of ice clouds for use in climate models. *J. Climate*, 22, 6287–6312, doi:10.1175/2009JCLI2844.1.
- Houze, R. A., P. V. Hobbs, P. H. Herzegh, and D. B. Parsons, 1979: Size distributions of precipitation particles in frontal clouds. J. Atmos. Sci., 36, 156–162, doi:10.1175/ 1520-0469(1979)036<0156:SDOPPI>2.0.CO;2.
- Kokhanovsky, A., 2004: Optical properties of terrestrial clouds. *Earth-Sci. Rev.*, 64, 189–241, doi:10.1016/S0012-8252(03)00042-4.
- Korolev, A. V., E. F. Emery, J. W. Strapp, S. G. Cober, G. A. Isaac, M. Wasey, and D. Marcotte, 2011: Small ice particles in tropospheric clouds: Fact or artifact? Airborne Icing Instrumentation Evaluation experiment. *Bull. Amer. Meteor. Soc.*, **92**, 967–973, doi:10.1175/2010BAMS3141.1.
- , —, —, , and —, 2013: Quantification of the effects of shattering on airborne ice particle measurements.
 J. Atmos. Oceanic Technol., 30, 2527–2553, doi:10.1175/JTECH-D-13-00115.1.
- Kristjánsson, J. E., J. M. Edwards, and D. L. Mitchell, 2000: Impact of a new scheme for optical properties of ice crystals on climates of two GCMs. J. Geophys. Res., 105, 10063–10079, doi:10.1029/2000JD900015.
- Lindner, T. H., and J. Li, 2000: Parameterization of the optical properties for water clouds in the infrared. *J. Climate*, 13, 1797–1805, doi:10.1175/1520-0442(2000)013<1797: POTOPF>2.0.CO;2.
- Loeb, N. G., B. A. Wielicki, D. R. Doelling, G. L. Smith, D. F. Keyes, S. Kato, N. Manalo-Smith, and T. Wong, 2009: Toward optimal closure of the earth's top-of-atmosphere radiation budget. J. Climate, 22, 748–766, doi:10.1175/ 2008JCLI2637.1.
- Lynch, D. K., K. Sassen, D. O'C. Starr, and G. L. Stephens, Eds., 2002: Cirrus. Oxford University Press, 504 pp.

- Manners, J., J.-C. Thelen, J. Petch, P. Hill, and J. M. Edwards, 2009: Two fast radiative transfer methods to improve the temporal sampling of clouds in numerical weather prediction and climate models. *Quart. J. Roy. Meteor. Soc.*, **135**, 457–468, doi:10.1002/qj.385.
- McFarquhar, G. M., and A. J. Heymsfield, 1998: The definition and significance of an effective radius for ice clouds. J. Atmos. Sci., 55, 2039–2052, doi:10.1175/1520-0469(1998)055<2039: TDASOA>2.0.CO;2.
- —, S. Iacobellis, and R. C. J. Somerville, 2003: SCM simulations of tropical ice clouds using observationally based parameterizations of microphysics. J. Climate, 16, 1643–1664, doi:10.1175/ 1520-0442(2003)016<1643:SSOTIC>2.0.CO;2.
- —, J. Um, and R. Jackson, 2013: Small cloud particle shapes in mixed-phase clouds. J. Appl. Meteor. Climatol., 52, 1277–1293, doi:10.1175/JAMC-D-12-0114.1.
- Mitchell, D. L., 2002: Effective diameter in radiation transfer: General definition, applications, and limitations. J. Atmos. Sci., 59, 2330–2346, doi:10.1175/1520-0469(2002)059<2330: EDIRTG>2.0.CO;2.
- —, A. Macke, and Y. Liu, 1996: Modeling cirrus clouds. Part II: Treatment of radiative properties. *J. Atmos. Sci.*, 53, 2967–2988, doi:10.1175/1520-0469(1996)053<2967: MCCPIT>2.0.CO;2.
- —, P. Rasch, D. Ivanova, G. M. McFarquhar, and T. Nousiainen, 2008: Impact of small ice crystal assumptions on ice sedimentation rates in cirrus clouds and GCM simulations. *Geophys. Res. Lett.*, **35**, L09806, doi:10.1029/ 2008GL033552.
- —, R. P. d'Entremont, and R. P. Lawson, 2010: Inferring cirrus size distributions through satellite remote sensing and microphysical databases. J. Atmos. Sci., 67, 1106–1125, doi:10.1175/ 2009JAS3150.1.
- —, R. P. Lawson, and B. Baker, 2011a: Understanding effective diameter and its application to terrestrial radiation in ice clouds. *Atmos. Chem. Phys.*, **11**, 3417–3429, doi:10.5194/ acp-11-3417-2011.
- —, S. Mishra, and R. P. Lawson, 2011b: Representing the ice fall speed in climate models: Results from Tropical Composition, Cloud and Climate Coupling (TC4) and the Indirect and Semi-Direct Aerosol Campaign (ISDAC). J. Geophys. Res., 116, D00T03, doi:10.1029/2010JD015433.
- Stephens, G. L., and Coauthors, 2012: An update on Earth's energy balance in light of the latest global observations. *Nat. Geosci.*, 5, 691–696, doi:10.1038/ngeo1580.

- Stocker, T. F., and Coauthors, 2013: Climate Change 2013: The Physical Science Basis. Cambridge University Press, 1535 pp. [Available online at www.climatechange2013.org/images/ report/WG1AR5_ALL_FINAL.pdf.]
- van de Hulst, H. C., 1957: Light Scattering by Small Particles. Wiley, 470 pp.
- Waliser, D., and Coauthors, 2009: Cloud ice: A climate model challenge with signs and expectations of progress. J. Geophys. Res., 114, D00A21, doi:10.1029/2008JD010015.
- Walters, D., and Coauthors, 2011: The Met Office Unified Model Global Atmosphere 3.0/3.1 and JULES Global Land 3.0/3.1 configurations. *Geosci. Model Dev.*, 4, 919–941, doi:10.5194/ gmd-4-919-2011.
- —, and Coauthors, 2013: The Met Office Unified Model Global Atmosphere 4.0 and JULES Global Land 4.0 configurations. *Geosci. Model Dev. Discuss.*, 6, 2813–2881, doi:10.5194/ gmdd-6-2813-2013.
- Warren, S. G., and R. E. Brandt, 2008: Optical constants of ice from the ultraviolet to the microwave: A revised compilation. J. Geophys. Res., 113, D14220, doi:10.1029/ 2007JD009744.
- Westbrook, C. D., R. C. Ball, P. R. Field, and A. J. Heymsfield, 2004: Theory of growth by differential sedimentation, with application to snowflake formation. *Phys. Rev.*, **70E**, 021403, doi:10.1103/PhysRevE.70.021403.
- Williams, K. D., and M. J. Webb, 2009: A quantitative performance assessment of cloud regimes in climate models. *Climate Dyn.*, 33, 141–157, doi:10.1007/s00382-008-0443-1.
- Wood, N., and Coauthors, 2014: An inherently mass-conserving semi-implicit semi-Langrangian discretization of the deepatmosphere global non-hydrostatic equations. *Quart. J. Roy. Meteor. Soc.*, 140, 1505–1520, doi:10.1002/qj.2235.
- Wu, D. L., and Coauthors, 2009: Comparisons of global cloud ice from MLS, CloudSat, and correlative data sets. J. Geophys. Res., 114, D00A24, doi:10.1029/2008JD009946.
- Wyser, K., and P. Yang, 1998: Average ice crystal size and bulk short-wave single-scattering properties of cirrus clouds. *Atmos. Res.*, 49, 315–335, doi:10.1016/S0169-8095(98)00083-0.
- Yang, P., and K.-N. Liou, 1998: Single-scattering properties of complex ice crystals in terrestrial atmosphere. *Contrib. Atmos. Phys.*, **71**, 223–248.
- Yi, B., P. Yang, B. A. Baum, T. L'Ecuyer, L. Oreopoulos, E. J. Mlawer, A. J. Heymsfield, and K.-N. Liou, 2013: Influence of ice particle surface roughness on the global cloud radiative effect. *J. Atmos. Sci.*, **70**, 2794–2807, doi:10.1175/JAS-D-13-020.1.